Author's response to reviews

Title: Postprandial glucose-lowering effects of fermented red ginseng in subjects with impaired fasting glucose or type 2 diabetes: A randomized, double-blind, placebo-controlled clinical trial

Authors:

Mi-Ra Oh (omr@ctcf2.com)
Soo-Hyun Park (psh@ctcf2.com)
Sun-Young Kim (Kimsy@ctcr2.com)
Hyang-Im Back (hyangim100@gmail.com)
Min-Gul Kim (mgkim@jbctc.org)
Ji-Young Jeon (jyjeon@jbctc.org)
Ki-Chan Ha (omphalos@chonbuk.ac.kr)
Won-Taek Na (wonpharm@hanmail.net)
Youn-Soo Cha (cha8@jbnu.ac.kr)
Byung-Hyun Park (bhpark@jbnu.ac.kr)
Tae-sun Park (pts@jbnu.ac.kr)
Soo-Wan Chae (soowan@jbnu.ac.kr)

Version: 3
Date: 22 October 2013

Author's response to reviews: see over
September 27, 2013
Managing Editor,
BMC Complementary and Alternative Medicine

Dear Managing Editor;

I am submitting herewith a full-length manuscript entitled “Postprandial glucose lowering effects of fermented red ginseng in subjects with impaired fasting glucose or type 2 diabetes: A randomized, double-blind, placebo-controlled clinical trial” by Mi-Ra Oh et al. In this study, we investigated the glucose lowering effects of fermented red ginseng on subjects with impaired fasting glucose or diabetes. Our results suggest that supplementation with fermented red ginseng resulted in a significant reduction of postprandial glucose level. This manuscript has neither been published nor is currently under consideration for publication either in whole or in part, by any other journal. All authors have agreed to be so listed, and have seen and approved the manuscript for publication in the “BMC Complementary and Alternative Medicine”.

I am looking forward to hearing you soon.

Sincerely Yours,

Soo-Wan Chae, MD, PhD
Clinical Trial Center,
Chonbuk National University Hospital,
Jeonju, Jeonbuk, 561-756, Korea
Tel: 82-63-250-2348
Fax: 82-63-250-2349
Email:soowan@jbnu.ac.kr